

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL103531

Key Points:

- CMIP6 models perform better than CMIP5 models on average
- Southern Hemisphere model ranking similar to Northern Hemisphere ranking
- More complex model versions perform similar to less complex ones

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. Brands, brandssf@ifca.unican.es

Citation:

Brands, S., Fernández-Granja, J. A., Bedia, J., Casanueva, A., & Fernández, J. (2023). A global climate model performance atlas for the Southern Hemisphere extratropics based on regional atmospheric circulation patterns. *Geophysical Research Letters*, 50, e2023GL103531. https://doi. org/10.1029/2023GL103531

Received 3 MAR 2023 Accepted 4 MAY 2023

© 2023. The Authors. Geophysical Research Letters published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

A Global Climate Model Performance Atlas for the Southern Hemisphere Extratropics Based on Regional Atmospheric Circulation Patterns

S. Brands^{1,2} D, J. A. Fernández-Granja¹ , J. Bedia^{3,4} , A. Casanueva^{3,4} , and J. Fernández¹

¹Instituto de Física de Cantabria (IFCA), CSIC-Universidad de Cantabria, Santander, Spain, ²MeteoGalicia, Consellería de Medio Ambiente, Territorio y Vivienda - Xunta de Galicia, Santiago de Compostela, Spain, ³Dept. Matemática Aplicada y Ciencias de la Computación (MACC), Universidad de Cantabria, Santander, Spain, ⁴Grupo de Meteorología y Computación, Universidad de Cantabria, Spain

Abstract The performance of 61 global climate models participating in CMIP5 and 6 is evaluated for the Southern Hemisphere extratropics in terms of typical regional-scale atmospheric circulation patterns. These patterns are known to be linked with a number of key variables in atmospheric physics and chemistry and provide an overarching concept for model evaluation. First, hemispheric-wide error and ranking maps are provided for each model and regional details are described. Then, the results are compared with those obtained in a companion study for the Northern Hemisphere. For most models, the average error magnitude and ranking position is similar on both hemispheres, ruling out systematic tuning toward either of the two. CMIP6 models perform better on average than CMIP5 models and the interactive simulation of more climate system components does not deteriorate the results for most model families. Better performance is associated with higher resolution in the atmosphere, following a non-linear relationship.

Plain Language Summary This letter provides a survey on the capability of global climate models to reproduce the regional atmospheric circulation in the Southern Hemisphere in present climate conditions. Climate models from the latest model generation are found to perform better on average than those of the previous generation and the obtained model ranking is similar to that found for the Northern Hemisphere in a companion study. While model performance is found to be generally unrelated to model complexity in terms of covered climate system components, better results are associated with higher model resolution in the atmosphere.

1. Introduction

The vast ocean and ice-sheet areas in the Southern Hemisphere (SH) extratropics are virtually uninhabited, but play a key role for the global climate system and are thus of paramount importance for mankind. In this context, the quasi-persistent circumpolar westerly winds blowing along the open sea channel in the mid-latitudes are of key relevance for several reasons. Partly offset by meso-scale ocean eddies tending to break up the intense ocean stratification, the westerlies drive the up-welling of carbon and nutrient-rich deep water and also force the Antarctic Circumpolar Current (ACC), which is embedded in the Meridional Overturning Circulation that in turn governs the low frequency variability of the global climate system (Abernathey et al., 2011; Hogg et al., 2017; Meredith et al., 2012; Speer & Marshall, 2012; Zhang et al., 2019). Commonly described by the Southern Annular Mode, also referred to as "Antarctic Oscillation" (Fogt & Marshall, 2020; Rogers & van Loon, 1982; Thompson & Wallace, 2000; Trenberth, 1979), the westerlies have shifted poleward during the last decades while, simultaneously, the Hadley Cell and associated large-scale subsidence in the sub-tropics have intensified (Fogt & Marshall, 2020; Nguyen et al., 2015; Thompson et al., 2000). Both anomalies are projected to magnify during the course of the 21st century in Global Climate Model (i.e., coupled General Circulation Model, GCM) experiments (Deng et al., 2022), leading to more frequent extreme events like, for example, droughts (Holgate et al., 2020) or sea-surface warming events (Duran et al., 2020), whose accumulated effects also alter the mass balance of the glaciers and ice-sheets in the SH.

While Patagonian glaciers are mainly affected by temperature and precipitation anomalies (Boex et al., 2013), Antarctic continental ice loss is mainly driven by melting into the Amundsen and Bellingshausen Seas (Hughes, 1981; Rignot et al., 2019). The aforementioned poleward shift of the westerlies has led to an enhanced

transport of relatively warm Circumpolar Deep Water, located at intermediate depths below the cold surface ocean layer, toward the continental shelf of the aforementioned sea areas (Steig et al., 2012), thereby thinning the ice shelves from below and melting the glaciers and ice streams at their ground lines. Subject to large uncertainties (Rignot et al., 2011), these processes contribute to global sea-level rise (Fox-Kemper et al., 2021).

Over glacial-to-interglacial cycles, the strength and position of the westerlies are also considered key to variations in the upwelling of carbon-rich Antarctic Bottom Water (AABW) reservoirs, associated with CO₂ degassing into the atmosphere (Sigman & Boyle, 2000; Speer & Marshall, 2012). There are indications that strong westerlies located near the Antarctic continent—well aligned with the ACC—, typically occur during warm, interglacial periods and enhance the aforementioned process leading to an increase in global CO₂ concentrations. Weaker westerlies located far away from the Antarctic continent and poorly aligned with the ACC are, in turn, currently discussed to be characteristic of cold, glacial periods. CO₂ degassing into the atmosphere would be reduced in this case, favoring a net carbon storage in the AABW (Gray et al., 2021; Toggweiler et al., 2006). Finally, AABW formation itself is also controlled by wind forcing, namely by southerly katabatic winds blowing down the Antarctic continent, pushing the sea-ice away from coast and thereby forming coastal polynyas (Tamura et al., 2008). In these ocean water areas surrounded by sea-ice, the nutrient-rich upwelled waters are subject to brine rejection during sea-ice formation that leads to increase in salinity. Sinking to the ocean bottom is the consequence, where "preformed" nutrients can thereby accumulate. AABW formation is particularly productive in the Weddell and Ross Seas and is subject to pronounced low-frequency variability (Hogg et al., 2017; Ito & Follows, 2005; Silvano et al., 2020).

These considerations show that the atmospheric circulation in the SH extratropics, even in confined and relatively small regions such as the aforementioned sea areas, are relevant for the entire climate system. Consequently, comprehensive GCMs now extensively used in climate research should perform well in this regard.

The present study evaluates a large multi-model ensemble from the Coupled Model Intercomparison Projects 5 and 6 (Eyring et al., 2016; Taylor et al., 2012) in terms of the models' capability to reproduce the climatological frequencies of typical and recurrent patterns of the regional atmospheric circulation in the SH extratropics. To this aim, the Lamb Weather Types method (LWT), also known as Jenkison-Collison circulation typing approach (Jenkinson & Collison, 1977; Jones et al., 1993; Lamb, 1972) has been recently extended for systematic use in the SH (Fernández-Granja et al., 2023) and is here applied to 61 GCMs from CMIP5 and 6, and to 3 distinct reference reanalyses. The circulation types obtained from this method are known to correlate with many key variables in atmospheric physics and chemistry and therefore constitute an overarching concept to describe regional-scale climate variability. The method is thus complementary to those operating on larger scales used in a previous study (Bracegirdle et al., 2020) and it is a direct answer to the downscaling community's claim for process-based GCM evaluation based on the regional atmospheric circulation (Maraun et al., 2017; Røste & Landgren, 2022), here tailored to the SH mid-to-high latitudes (Charles et al., 2020; Evans et al., 2021; Fita et al., 2017; Olson et al., 2016). Together with the respective assessment for the Northern Hemisphere (Brands, 2022b), this study completes the picture of GCM performance in terms of regional atmospheric circulation patterns in the extratropics. Possible model tuning issues to either of the two hemispheres are also discussed.

2. Data and Methods

The study relies on 6-hourly instantaneous sea-level pressure data from 61 different GCM configurations participating in CMIP5 and 6, all retrieved from the ESGF data portals. *Historical* experiments are evaluated and the individual ensemble members considered in the inter-model performance comparison, as well the Digital Object Identifiers of their reference articles, are listed in the *get_historical_metadata.py* function available from Brands, Tatebe, et al. (2023). It will be shown that the role of internal model variability does not lead to substantial changes in the results (see Section 3.1). Since several EC-Earth model versions were found to be favored when evaluated against ECMWF reanalyses (Dee et al., 2011; Hersbach et al., 2020) in the companion study conduced over the Northern Hemisphere (Brands, 2022b), the Japanese 55-year reanalysis (JRA-55) is here used as main reference data set for model evaluation (Kobayashi et al., 2015).

The Jenkinson-Collison circulation types constitute an objective classification method based on the subjective approach made by Lamb (1972). This technique, also known as "Lamb Weather Types" (Jones et al., 1993), groups an instantaneous SLP pattern centered at a given grid-box into 27 classes depending on the direction of

48007,

48007, 2023

1029/2023GL103531 by Csie

or) (Urici), Wiley Online

Librar

the geostrophic flow (or lack thereof) and the sign and strength of the vorticity. In addition to the purely cyclonic and anticyclonic types, there are 8 "purely directional" types—one for each of the 8 main cardinal directionsand 16 hybrid types characterized by a predominant flow from one of these directions combined with either cyclonic or anticyclonic conditions. A detailed description of the LWT method, including the extension to the SH relevant here, is provided in Fernández-Granja et al. (2023). The corresponding Python code is available from Brands (2022d). The LWT method is here applied in a rolling manner (Otero et al., 2017) looping through all boxes of a regular latitude-longitude grid covering a spatial domain extending from 30°S to 70°S with a 2.5° resolution. The considered time period is 1979–2005, for which data is available for all applied GCMs and reanalyses.

The LWT method is here said to be applicable for a given region if at least 20 types occur with a minimum relative frequency of 0.1% (i.e., n = 39 occurrences for 27 years and 6-hourly data) at the corresponding grid-box in the reference reanalysis (JRA-55). This criterion is fulfilled in virtually the entire study area. Moreover, to ensure that the regional-scale GCM ranking presented here is robust to a switch in the underlying reference reanalysis, ERA-Interim is evaluated against JRA-55 just as if it was another GCM and the obtained error is compared to the errors of the 61 authentic GCMs. If any of the considered GCMs is found to perform better than ERA-Interim at a given grid-box, this indicates large observational uncertainties due to sparse observations in the corresponding region, leading to an exclusion of the grid-box from further assessment. Figure 1 shows that virtually all grid-boxes along the Antarctic coastline and adjacent sea-areas covered by sea-ice during winter have to be excluded for this reason, the northern Antarctic Peninsula being the only exception.

For consistency with Brands (2022b) and Brands (2022c), the mean absolute error (MAE) of the climatological relative frequencies of the n = 27 LWTs is used as main error measure at the grid-box scale:

$$MAE = \frac{1}{n} \sum_{i=1}^{n} |m_i - o_i|$$
(1)

where m_i and o_i are the relative frequencies of the *i*th LWT (with i = 1, 2, ..., 27) from the GCM and reference reanalysis, respectively. Alternative error measures such as the Transition Probability Matrix Score (Fernandez-Granja et al., 2021) have been used as well, obtaining nearly identical results for the model ranking.

To explore the role of internal model variability due to initial conditions uncertainties, up to 18 distinct historical runs per GCM are evaluated for a subset of 13 GCMs, specified in Figure S3 in Supporting Information S1. Then, the GCM complexity score proposed in Brands (2022b) is used, which is proportional to the number of Earth system components taken into account in the GCM and gives more weight on simulated than on prescribed components. This score is put into relation with the spatial median model performance over the SH in order to explore whether the more complex GCMs perform better or worse than the less complex ones on average. Finally, the SH results are plotted against the NH results obtained from Brands (2022a) in order to detect possible tuning efforts to either of the two hemispheres. For a comparison on equal terms, the NH results were modified by also removing the regions prone to substantial reanalysis uncertainties, as described above.

3. Results

3.1. Regional Details

Figures 1 and 2 show the GCM ranking patterns based on the MAE for the 61 considered GCMs, with lower MAE values leading to better ranks and vice versa. The MAE values themselves are depicted in Figures S1 and S2 in Supporting Information S1. Hereafter, individual GCMs will be grouped according to the atmosphere general circulation model (AGCM) used therein (Brands, 2022c).

From these figures, it can be seen that the IFS AGCM family (i.e., all EC-Earth GCM versions) performs best overall, followed by the HadGAM/UM AGCM family comprising the ACCESS and HadGEM GCMs, as well as KACE1.0 G. All members of the HadGAM/UM family except ACCESS1.0 and HadGEM3-GC31-MM have similar error patterns, with larger errors in the Southern Ocean to the south, southwest and southeast of the Australian continent. ACCESS1.3, ACCESS-CM2, HadGEM3-GC31-MM and KACE1.0-G perform relatively poorly to the south and southwest of Cape of Good Hope and KACE1.0-G additionally performs badly off the east coast of South America. A large performance gain is observed from CSIRO-MK3.6 to the ACCESS GCM family, that is, from the former to the present GCM family developed by CSIRO (see also Figure 3).



Geophysical Research Letters



Figure 1. Ranking of the GCMs according to MAE defined in Equation 1, reference: JRA-55, 1979–2005, Part 1.



Geophysical Research Letters



Figure 2. As Figure 1, Part 2.



Geophysical Research Letters



Figure 3. Summary model performance plot. Columns are constructed upon the model-specific, point-wise error values over the Southern Hemisphere as depicted in Figures 1 and 2. The four additional boxplots depicted in light green were built upon the joint error samples of the more and the less complex GCMs used in CMIP5 and 6, respectively. Colors refer to research institutes as listed in the legend. The acronyms of the coupled models, as well as their participation in either CMIP5 or 6 (indicated by the final integer) are shown below the *X*-axis. Above this axis, the atmospheric component of each coupled model is shown in addition. Results are for the 1979–2005 period and w.r.t. JRA-55. AGCM abbreviations along the *X*-axis are as defined as follows: (1) MK3-AGCM, (2) GAMIL, (3) BCC-AGCM, (4) INM-AM, (5) CanAM4 and 6) GFS; the names of the remaining AGCMs are indicated in the figure.

The GAMIL AGCM family comprising the FGOALS-g2 and g3 GCMs performs overly poorest in this multi-model comparison. The ECHAM AGCM family, including all MPI-ESM versions, AWI-ESM-1-1-LR, NESM3 and CMCC-CM, performs slightly worse than the IFS and HadGAM/UM families, except for MPI-ESM1-2-HR and MPI-ESM-1-2-HAM performing almost equally well. A particularly poor model performance is observed for NESM3 along virtually the entire subtropics, extending to the mid-latitudes in the South Atlantic Ocean, and for AWI-ESM-1-1-LR over the eastern South Pacific and eastern South Atlantic. The CAM AGCM family comprises the largest number of GCMs (CMCC-CM2-SR5, CMCC-CM2-HR4, CMCC-ESM2, CCSM4, NorESM1-M, NorESM2-LM, NorESM2-MM, SAM0-UNICON, TaiESM1, BCC-CSM1.1 and BCC-CSM2-MR) and yields intermediate to unfavorable ranks in most regions. CMCC-CM, NorESM2-MM and SAM0-UNICON perform best in this family, yielding very good ranks in specific regions. CanESM2 comprises the CanAM4 AGCM that is not used in any other GCM and performs relatively poorly.

48007, 2023,

19448007, 2023

1029/2023GL103531 by Csie

Mayor) (Urici), Wiley Online

Library on [23/11/2023]

For the ARPEGE AGCM family shown in Figure 2 (CNRM-CM5, CNRM-CM6-1, CNRM-CM6-1-HR and CNRM-ESM2-1), the model versions used in CMIP6 (CNRM-CM6-1, CNRM-CM6-1-HR and CNRM-ESM2-1) perform worse than the well performing CMIP5 version CNRM-CM5. Surprisingly, this decrease in model performance is particularly pronounced in the high resolution (HR) version. IFS (EC-Earth2.3, EC-Earth3, EC-Earth3-Veg, EC-Earth3-Veg-LR, EC-Earth3-AerChem and EC-Earth3-CC) is the best performing model family in the present study, obtaining very good ranks over a large fraction of the domain. Model ranks worse than 40 are very rare, except for the ocean area to the south of Africa in EC-Earth-Veg-LR. The performance of the GFDL-AM AGCM family comprising GFDL-CM3, GFDL-CM4, GFDL-ESM2G, GFDL-ESM4 and KIOST-ESM is similar in magnitude to the ECHAM family, with best results overall for GFDL-ESM4. In case of the GISS-E2 AGCM family, the use of the Russel ocean model in GISS-E2-R leads to substantially better results than the use of the HYCOM model used in GISS-E2-H, the configuration of these two GCMs being otherwise equal (Schmidt et al., 2014), and a further performance increase is obtained by the CMIP6 version GISS-E2.1-G, comparable to that obtained for the ECHAM and GFDL-AM families mentioned above (see Figure 3). The most pronounced performance gain from CMIP5 to 6 is obtained for the LMDZ AGCM family (i.e., from IPSL-CM5A-LR and MR to IPSL-CM6A-LR), yielding a MAE level for IPSL-CM6A-LR comparable to that obtained for the ECHAM and GFDL-AM families. The MIROC-AM family is prone to very large performance differences from the better performing versions MIROC5 and 6 to the substantially worse performing versions MIROC-ESM and MIROC-ES2L, which are both more complex (see Figure 4d). The performance of the GSMUV family decreases substantially from CMIP5 to 6 (from MRI-ESM1 and MRI-ESM2.0) whereas that of the INM-AM family increases drastically (from INM-CM4 to INM-CM5). Finally, IITM-ESM is one of the worst performing GCMs considered here, with large differences in the results from one region to another.

3.2. Performance Summary and Comparison With the NH Results

In Figure 3, the hemispheric-wide MAE samples mapped in Figures S1 and S2 in Supporting Information S1 are displayed in a single boxplot. Each item describes the error distribution of a specific GCM in terms of the median (horizontal black line), interquartile range (IQR, box) and whiskers extending to the full range, except for the outliers lying beyond $1.5 \times IQR$ below and above the second and third quartile, respectively. The last four boxes, depicted in light green, are built upon the joint samples of the more and the less complex GCMs used in CMIP5 and 6, respectively (outliers are not shown for these samples). To this end, the GCMs are grouped into two classes according to their complexity score obtained from Brands (2022b). Those obtaining a score ≥ 14 are considered *more complex* and those obtaining a lower score *less complex*. Along the *x*-axis, the GCMs are grouped according to the AGCM family they belong to (Brands, 2022c). An alternative ordering according to the spatial median MAE in ascending order is provided in Figure S4 in Supporting Information S1.

For both complexity classes, the models used in CMIP6 perform better on average than those used in CMIP5 (see the last four columns in Figure 3). The largest performance gains from CMIP5 to 6 are obtained for the FGOALS and IPSL GCMs. However, a performance loss is obtained for 4 GCM groups—ACCESS, CMCC-CM, CNRM-CM and MRI-ESM—, which are two groups more than for the NH results obtained in Brands (2022b). Figure S3 in Supporting Information S1 shows that internal model variability does not substantially change the aforementioned results.

A comparison between the areal median performance in the SH versus NH is provided in Figures 4a and 4b. Overall, GCM performance is better in the SH than in the NH (panel a), which may be simply due to the fact that GCMs tend to perform better over the ocean than over land (Brands, 2022b), the ocean area being much larger in the SH. A close correspondence is obtained for the median error samples of the two hemispheres, particularly if they are log-transformed and standardized separately in order to remove systematic differences in their hemisphere-specific shape, magnitude and dispersion (panel b). Largest deviations from the diagonal are obtained for CNRM-CM6-1-HR, MRI-ESM2, CMCC-CM2-SR5, CMCC-ESM2 and HadGEM3-GC31-MM, performing better over the NH, and for CSIRO-MK3.6, KIOST-ESM, GISS-E2-1-G, MPI-ESM1.2-HAM and INM-CM5, performing better over the SH.

A significant non-linear relationship is obtained between the resolution of the AGCM—here described by the number of grid-points constituting the global 3-dimensional mesh (longitudes × latitudes × vertical layers)—, and the median model performance, obtaining a Spearman correlation coefficient (*rs*) of -0.49. Higher resolution is associated with better performance, particularly above a threshold of approximately 1.8×10^7 grid-boxes. Note that CNRM-CM6-1-HR and CNRM-ESM2-1 are not shown in Figure 4c because they are out-of-scale due to their very high resolution. Interestingly, the corresponding link with the 3D resolution of the ocean sub-model is weak (rs = -0.29), yet significant at a test level of 5%.

7 of 12



Figure 4. (a) Median model performance per GCM over the SH versus NH; (b) As a, but for log-transformed and standardized data; (c) 3D mesh size of the AGCM versus median model performance of the GCM in the SH and (d) Model complexity score proposed by Brands (2022b) versus median model performance over the SH. AGCM families are indicated as follows: 1) HadGEM/UM, 2) GAMIL, 3) ECHAM, 4) CAM, 5) BCC-AGCM, 6) ARPEGE, 7) IFS, 8) GFDL-AM, 9) GISS-E2, 10) LMDZ, 11) MIROC/CCSR-AGCM, 12) GSMUV, 13) INM-AM.



19448007, 2023,

10, Downloaded

Finally, median model performance over the SH is generally not associated with model complexity (r = -0.01) and, for most model families, the more complex versions perform at least equally well than the less complex ones (see Figure 4d). The MIROC-AGCM family is an exception in this sense, since the more complex model versions MIROC-ESM and MIROC-ES2L, probably due to their low horizontal resolution in the atmosphere (T42) (Brands, Tatebe, et al., 2023), perform substantially worse than the less complex versions MIROC5 and 6 (T85).

4. Conclusions

In the present study, 61 different GCMs from CMIP5 and 6 have been evaluated in the SH extratropics excluding Antarctica, focusing on the models' ability to reproduce the climatological frequency of the 27 Lamb Weather Types, known to be associated with many environmental variables and thus constituting a overarching concept to regional-scale climate variability.

While all of the model families performing poorly in CMIP5 have improved considerably in CMIP6, most of the families already performing well in CMIP5 have suffered a slight performance loss. For most model families, the spatial average performance for the SH is similar to the NH (Brands, 2022b), suggesting that systematic model tuning to either of the two hemispheres can be ruled out in general terms. For a small number of specific GCMs, however, substantial performance differences are obtained from one hemisphere to another and the reasons for this should be assessed in future studies. Whereas a higher resolution in the atmospheric sub-model of the considered GCMs is found to be associated with better performance, following an exponentially decreasing relationship, GCM complexity as defined in Brands (2022b) is generally unrelated to performance, except for the MIROC-AGCM family, whose more complex versions perform worse than the less complex ones over the SH. This is a promising result since the more complex models are also prone to more error sources. It is also an argument for the use of the more complex models, as they provide a more complete picture of the feedback processes governing the climate system (Döscher et al., 2021; Dunne et al., 2020; Séférian et al., 2019). The here obtained results are expected to be of value for regional climate studies in the Southern Hemisphere involving GCMs, as well as for the model development teams themselves.

Data Availability Statement

Figures S1 to S4 are contained in Supporting Information S1 file to this article, available from GRL's homepage. The Python source code underlying this study and the GCM metadata archive get_historitcal_metadata.py are publicly available from Brands (2022d) and Brands, Tatebe, et al. (2023), respectively, and the LWT data set for the considered GCMs and reanalyses can be retrieved from Brands, Fernández-Granja et al. (2023a). Additional auxiliary material containing (a) separate pdf files for each error and ranking map, (b) netCDF files containing grid-box-scale GCM errors and (c) summary csv files listing the model complexity score from Brands (2022b), as well as the spatial median performance over the SH domain for each GCM, can be retrieved from Brands, Fernández-Granja et al. (2023b). The reference articles for the GCMs and reanalyses considered here are: Bentsen et al. (2013); Bi et al. (2013, 2020); Boucher et al. (2020); Cao et al. (2018); Cherchi et al. (2019); Chylek et al. (2011); Collins et al. (2011); Dee et al. (2011); Döscher et al. (2021); Dufresne et al. (2013); Dunne et al. (2012, 2020); Gent et al. (2011); Giorgetta et al. (2013); Griffies et al. (2011); Hajima et al. (2020); Hazeleger et al. (2011); Held et al. (2019); Hersbach et al. (2020); Kelley et al. (2020); Kobayashi et al. (2015); J. Lee et al. (2019); W.-L. Lee et al. (2020); Li et al. (2013, 2020); Martin et al. (2011); Mauritsen et al. (2019); Swapna et al. (2015); Pak et al. (2021); Park et al. (2019); Roberts et al. (2019); Scoccimarro et al. (2011); Schmidt et al. (2014); Séférian et al. (2019); Seland et al. (2020); Semmler et al. (2020); Tatebe et al. (2019); Tegen et al. (2019); Voldoire et al. (2013, 2019); Volodin et al. (2010, 2017); M. Watanabe et al. (2010); S. Watanabe et al. (2011); Wu et al. (2014, 2019); Yukimoto et al. (2011, 2019); Ziehn et al. (2020).

References

Abernathey, R., Marshall, J., & Ferreira, D. (2011). The dependence of Southern Ocean meridional overturning on wind stress. *Journal of Physical Oceanography*, 41(12), 2261–2278. https://doi.org/10.1175/JPO-D-11-023.1

Bentsen, M., Bethke, I., Debernard, J. B., Iversen, T., Kirkevåg, A., Seland, Ø., et al. (2013). The Norwegian Earth System Model, NorESM1-M – Part 1: Description and basic evaluation of the physical climate. *Geoscientific Model Development*, 6(3), 687–720. https://doi.org/10.5194/ gmd-6-687-2013

the European Commission - Next-GenerationEU (Regulation EU 2020/2094), through CSIC's Interdisciplinary Thematic Platform Clima (PTI Clima)/Development of Operational Climate Services and by the I+D+i project CORDyS (PID2020-116595RB-I00), funded by MCIN/ AEI/10.13039/501100011033. J.A.F. acknowledges support from project ATLAS (PID2019-111481RB-I00), grant PRE2020-094728 funded by MCIN/ AEI/10.13039/501100011033 and ESF investing in your future. A.C. acknowledges support from Project COMPOUND (TED2021-131334A-I00) funded by MCIN/AEI/10.13039/501100011033 and by the European Union NextGenerationEU/PRTR. S.B. would like to thank CESGA and AMTEGA for providing computational resources. The authors acknowledge the public availability of the CMIP datasets via the ESGF data portals, as well the free distribution of the ECMWF and JMA reanalysis products.

19448007, 2023,

1029/2023GL103531 by Csic Orga

Online

Library on [23/11/2023]

- Bi, D., Dix, M., Marsland, S., O'Farrell, S., Sullivan, A., Bodman, R., et al. (2020). Configuration and spin-up of ACCESS-CM2, the new generation Australian community climate and Earth system simulator coupled model. *Journal of Southern Hemisphere Earth Systems Science*, 70(1), 225–251. https://doi.org/10.1071/ES19040
- Bi, D., Dix, M., Marsland, S. J., O'Farrell, S., Rashid, H., Uotila, P., et al. (2013). The ACCESS coupled model: Description, control climate and evaluation. *Australian Meteorological and Oceanographic Journal*, 63(1), 41–64. https://doi.org/10.22499/2.6301.004
- Boex, J., Fogwill, C., Harrison, S., Glasser, N., Hein, A., Schnabel, C., & Xu, S. (2013). Rapid thinning of the late Pleistocene Patagonian ice sheet followed migration of the southern westerlies. *Scientific Reports*, 3(1), 2118. https://doi.org/10.1038/srep02118
- Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov, V., et al. (2020). Presentation and evaluation of the IPSL-CM6A-LR climate model. *Journal of Advances in Modeling Earth Systems*, 12(7), e2019MS002010. https://doi.org/10.1029/2019MS002010
- Bracegirdle, T. J., Holmes, C. R., Hosking, J. S., Marshall, G. J., Osman, M., Patterson, M., & Rackow, T. (2020). Improvements in circumpolar Southern Hemisphere extratropical atmospheric circulation in CMIP6 compared to CMIP5. *Earth and Space Science*, 7(6), e2019EA001065. https://doi.org/10.1029/2019EA001065
- Brands, S. (2022a). A circulation-based performance atlas of the CMIP5 and 6 models for regional climate studies in the northern hemisphere [dataset]. Zenodo. https://doi.org/10.5281/zenodo.4452080
- Brands, S. (2022b). A circulation-based performance atlas of the CMIP5 and 6 models for regional climate studies in the Northern Hemisphere mid-to-high latitudes. *Geoscientific Model Development*, *15*(4), 1375–1411. https://doi.org/10.5194/gmd-15-1375-2022
- Brands, S. (2022c). Common error patterns in the regional atmospheric circulation simulated by the CMIP multi-model ensemble. *Geophysical Research Letters*, 49(23), e2022GL101446. https://doi.org/10.1029/2022GL101446
- Brands, S. (2022d). Python code to calculate Lamb circulation types for the Northern Hemisphere derived from historical CMIP simulations and reanalysis data [code]. Zenodo. https://doi.org/10.5281/zenodo.4555367
- Brands, S., Fernández-Granja, J. A., Bedia, J., Casanueva, A., & Fernández, J. (2023a). Southern Hemisphere Lamb Weather Types from historical GCM experiments and various reanalyses (1.0) [dataset]. Zenodo. https://doi.org/10.5281/zenodo.7612988
- Brands, S., Fernández-Granja, J. A., Bedia, J., Casanueva, A., & Fernández, J. (2023b). Auxiliary online material to Brands et al. (2023): A global climate model performance atlas for the southern hemisphere extratropics based on regional atmospheric circulation patterns. *Figshare*. https://doi.org/10.6084/m9.figshare.22193443.v1
- Brands, S., Tatebe, H., Danek, C., Fernández, J., Swart, N., Volodin, E., et al. (2023). GCM metadata archive get_historical_metadata.py (v1.1). Zenodo. https://doi.org/10.5281/zenodo.7715383
- Cao, J., Wang, B., Yang, Y.-M., Ma, L., Li, J., Sun, B., et al. (2018). The NUIST Earth system model (NESM) version 3: Description and preliminary evaluation. Geoscientific Model Development, 11(7), 2975–2993. https://doi.org/10.5194/gmd-11-2975-2018
- Charles, S. P., Chiew, F. H. S., Potter, N. J., Zheng, H., Fu, G., & Zhang, L. (2020). Impact of downscaled rainfall biases on projected runoff changes. *Hydrology and Earth System Sciences*, 24(6), 2981–2997. https://doi.org/10.5194/hess-24-2981-2020
- Cherchi, A., Fogli, P. G., Lovato, T., Peano, D., Iovino, D., Gualdi, S., et al. (2019). Global mean climate and main patterns of variability in the CMCC-CM2 coupled model. *Journal of Advances in Modeling Earth Systems*, 11(1), 185–209. https://doi.org/10.1029/2018MS001369
- Chylek, P., Li, J., Dubey, M. K., Wang, M., & Lesins, G. (2011). Observed and model simulated 20th century arctic temperature variability: Canadian Earth system model CanESM2. Atmospheric Chemistry and Physics Discussions, 11, 22893–22907. https://doi.org/10.5194/ acpd-11-22893-2011
- Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Halloran, P., Hinton, T., et al. (2011). Development and evaluation of an Earth-system model HadGEM2. *Geoscientific Model Development*, 4(4), 1051–1075. https://doi.org/10.5194/gmd-4-1051-2011
- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137(656), 553–597. https://doi. org/10.1002/qj.828
- Deng, K., Azorin-Molina, C., Yang, S., Hu, C., Zhang, G., Minola, L., & Chen, D. (2022). Changes of Southern Hemisphere westerlies in the future warming climate. Atmospheric Research, 270, 106040. https://doi.org/10.1016/j.atmosres.2022.106040
- Döscher, R., Acosta, M., Alessandri, A., Anthoni, P., Arneth, A., Arsouze, T., et al. (2021). The EC-Earth3 Earth system model for the coupled model intercomparison project 6. *Geoscientific Model Development Discussions*, 1–90. https://doi.org/10.5194/gmd-2020-446
- Dufresne, J.-L., Foujols, M.-A., Denvil, S., Caubel, A., Marti, O., Aumont, O., et al. (2013). Climate change projections using the IPSL-CM5 Earth system model: From CMIP3 to CMIP5. *Climate Dynamics*, 40(9–10), 2123–2165. https://doi.org/10.1007/s00382-012-1636-1
- Dunne, J. P., Horowitz, L. W., Adcroft, A. J., Ginoux, P., Held, I. M., John, J. G., et al. (2020). The GFDL Earth system model version 4.1 (GFDL-ESM 4.1): Overall coupled model description and simulation characteristics. *Journal of Advances in Modeling Earth Systems*, 12(11), e2019MS002015. https://doi.org/10.1029/2019MS002015
- Dunne, J. P., John, J. G., Adcroft, A. J., Griffies, S. M., Hallberg, R. W., Shevliakova, E., et al. (2012). GFDL's ESM2 global coupled climate-carbon Earth system models. Part I: Physical formulation and baseline simulation characteristics. *Journal of Climate*, 25(19), 6646–6665. https://doi. org/10.1175/JCLI-D-11-00560.1
- Duran, E. R., England, M. H., & Spence, P. (2020). Surface ocean warming around Australia driven by interannual variability and long-term trends in Southern Hemisphere westerlies. *Geophysical Research Letters*, 47(9), e2019GL086605. https://doi.org/10.1029/2019GL086605
- Evans, J., Virgilio, G., Hirsch, A., Hoffmann, P., Remedio, A. R., Ji, F., et al. (2021). The CORDEX-Australasia ensemble: Evaluation and future projections. *Climate Dynamics*, 57(5–6), 1385–1401. https://doi.org/10.1007/s00382-020-05459-0
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model intercomparison project phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Fernández-Granja, J. A., Brands, S., Bedia, J., Casanueva, A., & Fernández, J. (2023). Exploring the limits of the Jenkinson–Collison weather types classification scheme: A global assessment based on various reanalyses. *Climate Dynamics*. https://doi.org/10.1007/s00382-022-06658-7
- Fernandez-Granja, J. A., Casanueva, A., Bedia, J., & Fernández, J. (2021). Improved atmospheric circulation over Europe by the new generation of CMIP6 Earth System Models. *Climate Dynamics*, 56(11–12), 3527–3540. https://doi.org/10.1007/s00382-021-05652-9
- Fita, L., Evans, J., Argueso, D., King, A., & Liu, Y. (2017). Evaluation of the regional climate response in Australia to large-scale climate modes in the historical NARCliM simulations. *Climate Dynamics*, 49(7–8), 1–15. https://doi.org/10.1007/s00382-016-3484-x
- Fogt, R. L., & Marshall, G. J. (2020). The southern annular mode: Variability, trends, and climate impacts across the southern hemisphere. WIREs Climate Change, 11(4), e652. https://doi.org/10.1002/wcc.652
- Fox-Kemper, B., Hewitt, H., Xiao, C., Aoalgeirsdottir, G., Drijfhout, S. S., Edwards, T., et al. (2021). Climate change 2021: The physical science basis. Contribution of working Group I to the sixth assessment report of the intergovernmental panel on climate change. In V. Masson-Delmotte, et al. (Eds.), (pp. 1211–1362). Cambridge University Press. https://doi.org/10.1017/9781009157896.011

- Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., et al. (2011). The community climate system model version 4. *Journal of Climate*, 24(19), 4973–4991. https://doi.org/10.1175/2011JCLI4083.1
- Giorgetta, M. A., Jungclaus, J., Reick, C. H., Legutke, S., Bader, J., Böttinger, M., et al. (2013). Climate and carbon cycle changes from 1850 to 2100 in MPI-ESM simulations for the coupled model intercomparison project phase 5. *Journal of Advances in Modeling Earth Systems*, 5(3), 572–597. https://doi.org/10.1002/jame.20038
- Gray, W., de Lavergne, C., Jnglin Wills, R., Menviel, L., Spence, P., Holzer, M., et al. (2021). Poleward shift in the Southern Hemisphere westerly winds synchronous with the deglacial rise in CO₂. *Research Square*, 1–50. https://doi.org/10.21203/rs.3.rs-404786/v1
- Griffies, S., Winton, M., Donner, L., Horowitz, L., Downes, S., Farneti, R., et al. (2011). The GFDL-CM3 coupled climate model: Characteristics of the ocean and sea ice simulations. *Journal of Climate*, 24(13), 3520–3544. https://doi.org/10.1175/2011JCL13964.1
- Hajima, T., Watanabe, M., Yamamoto, A., Tatebe, H., Noguchi, M. A., Abe, M., et al. (2020). Development of the MIROC-ES2L Earth system model and the evaluation of biogeochemical processes and feedbacks. *Geoscientific Model Development*, 13(5), 2197–2244. https://doi. org/10.5194/gmd-13-2197-2020
- Hazeleger, W., Wang, X., Severijns, C., Briceag, S., Bintanja, R., Sterl, A., et al. (2011). Ec-Earth v2.2: Description and validation of a new seamless Earth system prediction model. *Climate Dynamics*, 39(11), 1–19. https://doi.org/10.1007/s00382-011-1228-5
- Held, I. M., Guo, H., Adcroft, A., Dunne, J. P., Horowitz, L. W., Krasting, J., et al. (2019). Structure and performance of GFDL's CM4.0 climate model. *Journal of Advances in Modeling Earth Systems*, 11(11), 3691–3727. https://doi.org/10.1029/2019MS001829
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Hogg, A. M., Spence, P., Saenko, O. A., & Downes, S. M. (2017). The energetics of Southern Ocean upwelling. Journal of Physical Oceanography, 47(1), 135–153. https://doi.org/10.1175/JPO-D-16-0176.1
- Holgate, C. M., Van Dijk, A. I. J. M., Evans, J. P., & Pitman, A. J. (2020). Local and remote drivers of southeast Australian drought. *Geophysical Research Letters*, 47(18), e2020GL090238. https://doi.org/10.1029/2020GL090238
- Hughes, T. J. (1981). The weak underbelly of the West Antarctic ice sheet. Journal of Glaciology, 27(97), 518–525. https://doi.org/10.3189/ S002214300001159X
- Ito, T., & Follows, M. (2005). Preformed phosphate, soft tissue pump and atmospheric CO2. Journal of Marine Research, 63(4), 813–839. https:// doi.org/10.1357/0022240054663231
- Jenkinson, A., & Collison, F. (1977). An initial climatology of gales over the North Sea Tech. Rep. No. 18 (Vol. 62). Meterological Office. Jones, P. D., Hulme, M., & Briffa, K. R. (1993). A comparison of Lamb circulation types with an objective classification scheme. International Journal of Climatology, 13(6), 655–663. https://doi.org/10.1002/joc.3370130606
- Kelley, M., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Ruedy, R., Russell, G. L., et al. (2020). GISS-E2.1: Configurations and climatology. Journal of Advances in Modeling Earth Systems, 12(8), e2019MS002025. https://doi.org/10.1029/2019MS002025
- Kobayashi, S., Ota, Y., Harada, Y., Ebita, A., Moriya, M., Onoda, H., et al. (2015). The JRA-55 Reanalysis: General specifications and basic characteristics. Journal of the Meteorological Society of Japan. Ser. II, 93(1), 5–48. https://doi.org/10.2151/jmsj.2015-001
- Lamb, H. (1972). British Isles weather types and a register of daily sequence of circulation patterns, 1861-1971 (Vol. 116, pp. 85). Geophysical Memoir. (HMSO).
- Lee, J., Kim, J., Sun, M.-A., Kim, B.-H., Moon, H., Sung, H. M., & Byun, Y.-H. (2019). Evaluation of the Korea meteorological administration advanced community Earth-system model (K-ACE). Asia-Pacific Journal of Atmospheric Sciences, 56(3), 381–395. https://doi.org/10.1007/ s13143-019-00144-7
- Lee, W.-L., Wang, Y.-C., Shiu, C.-J., Tsai, I., Tu, C.-Y., Lan, Y.-Y., et al. (2020). Taiwan Earth system model version 1: Description and evaluation of mean state. *Geoscientific Model Development*, 13(9), 3887–3904. https://doi.org/10.5194/gmd-13-3887-2020
- Li, L., Lin, P., Yu, Y.-Q., Zhou, T., Liu, L., Liu, J., et al. (2013). The flexible global ocean-atmosphere-land system model, grid-point version 2: FGOALS-g2. Advances in Atmospheric Sciences, 30(3), 543–560. https://doi.org/10.1007/s00376-012-2140-6
- Li, L., Yu, Y., Tang, Y., Lin, P., Xie, J., Song, M., et al. (2020). The flexible global ocean-atmosphere-land system model grid-point version 3 (FGOALS-g3): Description and evaluation. *Journal of Advances in Modeling Earth Systems*, 12(9), e2019MS002012. https://doi.org/10.1029/2019MS002012
- Maraun, D., Shepherd, T., Widmann, M., Zappa, G., Walton, D., Gutiérrez, J., et al. (2017). Towards process-informed bias correction of climate change simulations. *Nature Climate Change*, 7(11), 764–773. https://doi.org/10.1038/nclimate3418
- Martin, T. H. D. T. G. M., Bellouin, N., Collins, W. J., Culverwell, I. D., Halloran, P. R., Hardiman, S. C., et al. (2011). The HadGEM2 family of met office unified model climate configurations. *Geoscientific Model Development*, 4(3), 723–757. https://doi.org/10.5194/gmd-4-723-2011
- Mauritsen, T., Bader, J., Becker, T., Behrens, J., Bittner, M., Brokopf, R., et al. (2019). Developments in the MPI-M Earth system model version 1.2 (MPI-ESM1.2) and its response to increasing CO₂. Journal of Advances in Modeling Earth Systems, 11(4), 998–1038. https://doi. org/10.1029/2018MS001400
- Meredith, M. P., Garabato, A. C. N., Hogg, A. M., & Farneti, R. (2012). Sensitivity of the overturning circulation in the Southern Ocean to decadal changes in wind forcing. *Journal of Climate*, 25(1), 99–110. https://doi.org/10.1175/2011JCLI4204.1
- Nguyen, H., Lucas, C., Evans, A., Timbal, B., & Hanson, L. (2015). Expansion of the southern hemisphere Hadley cell in response to greenhouse gas forcing. *Journal of Climate*, 28(20), 8067–8077. https://doi.org/10.1175/JCLI-D-15-0139.1

Olson, R., Evans, J., Di Luca, A., & Argueso, D. (2016). The NARCliM project: Model agreement and significance of climate projections. *Climate Research*, 69(3), 209–227. https://doi.org/10.3354/cr01403

- Otero, N., Sillmann, J., & Butler, T. (2017). Assessment of an extended version of the Jenkinson-Collison classification on CMIP5 models over Europe. *Climate Dynamics*, 50(5–6), 1559–1579. https://doi.org/10.1007/s00382-017-3705-y
- Pak, G., Noh, Y., Lee, M.-I., Yeh, S.-W., Kim, D., Kim, S.-Y., et al. (2021). Korea institute of ocean science and technology Earth system model and its simulation characteristics. *Ocean Science Journal*, 56(1), 18–45. https://doi.org/10.1007/s12601-021-00001-7
- Park, S., Shin, J., Kim, S., Oh, E., & Kim, Y. (2019). Global climate simulated by the Seoul national university atmosphere Model version 0 with a unified convection scheme (SAM0-UNICON). *Journal of Climate*, 32(10), 2917–2949. https://doi.org/10.1175/JCLI-D-18-0796.1
- Rignot, E., Mouginot, J., Scheuchl, B., van den Broeke, M., van Wessem, M. J., & Morlighem, M. (2019). Four decades of Antarctic Ice Sheet mass balance from 1979-2017. Proceedings of the National Academy of Sciences, 116(4), 1095–1103. https://doi.org/10.1073/pnas.1812883116
- Rignot, E., Velicogna, I., van den Broeke, M. R., Monaghan, A., & Lenaerts, J. T. M. (2011). Acceleration of the contribution of the Greenland and Antarctic ice sheets to sea level rise. *Geophysical Research Letters*, 38(5), L05503. https://doi.org/10.1029/2011GL046583
- Roberts, M., Baker, A., Blockley, E., Calvert, D., Coward, A., Hewitt, H., et al. (2019). Description of the resolution hierarchy of the global coupled HadGEM3-GC3.1 model as used in CMIP6 HighResMIP experiments. *Geoscientific Model Development Discussions*, 1–47. https:// doi.org/10.5194/gmd-2019-148

- Rogers, J. C., & van Loon, H. (1982). Spatial variability of sea level pressure and 500 mb height anomalies over the Southern Hemisphere. Monthly Weather Review, 110(10), 1375–1392. https://doi.org/10.1175/1520-0493(1982)110(1375:SVOSLP)2.0.CO;2
- Røste, J., & Landgren, O. (2022). Impacts of dynamical downscaling on circulation type statistics in the Euro-CORDEX ensemble. *Climate Dynamics*, 59(7–8), 2445–2466. https://doi.org/10.1007/s00382-022-06219-y
- Schmidt, G. A., Kelley, M., Nazarenko, L., Ruedy, R., Russell, G. L., Aleinov, I., et al. (2014). Configuration and assessment of the GISS ModelE2 contributions to the CMIP5 archive. *Journal of Advances in Modeling Earth Systems*, 6(1), 141–184. https://doi.org/10.1002/2013MS000265
- Scoccimarro, E., Gualdi, S., Bellucci, A., Sanna, A., Giuseppe Fogli, P., Manzini, E., et al. (2011). Effects of tropical cyclones on ocean heat transport in a high-resolution coupled general circulation model. *Journal of Climate*, 24(16), 4368–4384. https://doi.org/10.1175/2011JCL14104.1
- Séférian, R., Nabat, P., Michou, M., Saint-Martin, D., Voldoire, A., Colin, J., et al. (2019). Evaluation of CNRM Earth system model, CNRM-ESM2-1: Role of Earth system processes in present-day and future climate. *Journal of Advances in Modeling Earth Systems*, 11(12), 4182–4227. https://doi.org/10.1029/2019MS001791
- Seland, Ø., Bentsen, M., Seland Graff, L., Olivié, D., Toniazzo, T., Gjermundsen, A., et al. (2020). The Norwegian Earth system model, NorESM2 – Evaluation of the CMIP6 DECK and historical simulations. *Geoscientific Model Development*, 1–68. https://doi.org/10.5194/gmd-2019-378 Semmler, T., Danilov, S., Gierz, P., Goessling, H. F., Hegewald, J., Hinrichs, C., et al. (2020). Simulations for CMIP6 with the AWI climate model
- AWI-CM-1-1. Journal of Advances in Modeling Earth Systems, 12(9), e2019MS002009. https://doi.org/10.1029/2019MS002009
- Sigman, D., & Boyle, E. (2000). Glacial/interglacial variations in atmospheric carbon dioxide. *Nature*, 407(6806), 859–869. https://doi.org/10.1038/35038000
- Silvano, A., Foppert, A., Rintoul, S., Holland, P., Tamura, T., Kimura, N., et al. (2020). Recent recovery of Antarctic bottom water formation in the Ross Sea driven by climate anomalies. *Nature Geoscience*, *13*(12), 1–7. https://doi.org/10.1038/s41561-020-00655-3
- Speer, K., & Marshall, J. (2012). Closure of the meridional overturning circulation through Southern Ocean upwelling. *Nature Geoscience*, 5(3), 171–180. https://doi.org/10.1038/ngeo1391
- Steig, E., Ding, Q., Battisti, D., & Jenkins, A. (2012). Tropical forcing of circumpolar deep water inflow and outlet glacier thinning in the Amundsen Sea Embayment, West Antarctica. Annals of Glaciology, 53(60), 19–28. https://doi.org/10.3189/2012AoG60A110
- Swapna, P., Koll, R., Aparna, K., Kulkarni, K., Ag, P., Ashok, K., et al. (2015). The IITM Earth System Model: Transformation of a seasonal prediction model to a long term climate model. *Bulletin of the American Meteorological Society*, 96(8), 1351–1367. https://doi.org/10.1175/ BAMS-D-13-00276.1
- Tamura, T., Ohshima, K. I., & Nihashi, S. (2008). Mapping of sea ice production for Antarctic coastal polynyas. *Geophysical Research Letters*, 35(7), L07606. https://doi.org/10.1029/2007GL032903
- Tatebe, H., Ogura, T., Nitta, T., Komuro, Y., Ogochi, K., Takemura, T., et al. (2019). Description and basic evaluation of simulated mean state, internal variability, and climate sensitivity in MIROC6. *Geoscientific Model Development*, 12(7), 2727–2765. https://doi.org/10.5194/ gmd-12-2727-2019
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Tegen, I., Neubauer, D., Ferrachat, S., Siegenthaler-Le Drian, C., Bey, I., Schutgens, N., et al. (2019). The global aerosol–climate model ECHAM6.3–HAM2.3 – Part 1: Aerosol evaluation. *Geoscientific Model Development*, 12(4), 1643–1677. https://doi.org/10.5194/ gmd-12-1643-2019
- Thompson, D. W. J., & Wallace, J. M. (2000). Annular modes in the extratropical circulation. Part I: Month-to-month variability. Journal of Climate, 13(5), 1000–1016. https://doi.org/10.1175/1520-0442(2000)013(1000:AMITEC)2.0.CO;2
- Thompson, D. W. J., Wallace, J. M., & Hegerl, G. C. (2000). Annular modes in the extratropical circulation. Part II: Trends. *Journal of Climate*, 13(5), 1018–1036. https://doi.org/10.1175/1520-0442(2000)013(1018:AMITEC)2.0.CO;2
- Toggweiler, J. R., Russell, J. L., & Carson, S. R. (2006). Midlatitude westerlies, atmospheric CO₂, and climate change during the ice ages. *Paleoceanography*, 21(2), PA2005. https://doi.org/10.1029/2005PA001154
- Trenberth, K. E. (1979). Interannual variability of the 500 mb zonal mean flow in the Southern Hemisphere. *Monthly Weather Review*, 107(11), 1515–1524. https://doi.org/10.1175/1520-0493(1979)107(1515:IVOTMZ)2.0.CO;2
- Voldoire, A., Saint-Martin, D., Sénési, S., Decharme, B., Alias, A., Chevallier, M., et al. (2019). Evaluation of CMIP6 DECK experiments with CNRM-CM6-1. Journal of Advances in Modeling Earth Systems, 11(7), 2177–2213. https://doi.org/10.1029/2019MS001683
- Voldoire, A., Sanchez-Gomez, E., Salas y Melia, D., Decharme, B., Cassou, C., Senesi, S., et al. (2013). The CNRM-CM5.1 global climate model: Description and basic evaluation. *Climate Dynamics*, 40(9–10), 2091–2121. https://doi.org/10.1007/s00382-011-1259-y
- Volodin, E., Diansky, N., & Gusev, A. (2010). Simulating present-day climate with the INMCM4.0 coupled model of the atmospheric and oceanic general circulations. *Izvestiya - Atmospheric and Oceanic Physics*, 46(4), 414–431. https://doi.org/10.1134/S000143381004002X
- Volodin, E., Mortikov, E., Kostrykin, S., Galin, V., Lykossov, V., Gritsun, A., et al. (2017). Simulation of the present-day climate with the climate model INMCM5. *Climate Dynamics*, 49(11–12), 3715–3734. https://doi.org/10.1007/s00382-017-3539-7
- Watanabe, M., Suzuki, T., O'ishi, R., Komuro, Y., Watanabe, S., Emori, S., et al. (2010). Improved climate simulation by MIROC5: Mean states, variability, and climate sensitivity. *Journal of Climate*, 23, 6312–6335. https://doi.org/10.1175/2010JCL13679.1
- Watanabe, S., Hajima, T., Sudo, K., Nagashima, T., Takemura, T., Okajima, H., et al. (2011). MIROC-ESM 2010: Model description and basic results of CMIP5-20c3m experiments. *Geoscientific Model Development*, 4, 845–872. https://doi.org/10.5194/gmd-4-845-2011
- Wu, T., Lu, Y., Fang, Y., Xin, X., Li, L., Li, W., et al. (2019). The Beijing climate center climate system model (BCC-CSM): The main progress from CMIP5 to CMIP6. *Geoscientific Model Development*, 12(4), 1573–1600. https://doi.org/10.5194/gmd-12-1573-2019
- Wu, T., Song, L., Li, W., Wang, Z., Zhang, H., Xin, X., et al. (2014). An overview of BCC climate system model development and application for climate change studies. Acta Meteorologica Sinica, 28(1), 34–56. https://doi.org/10.1007/s13351-014-3041-7
- Yukimoto, S., Kawai, H., Koshiro, T., Oshima, N., Yoshida, K., Urakawa, S., et al. (2019). The meteorological research institute Earth system model version 2.0, MRI-ESM2.0: Description and basic evaluation of the physical component. *Journal of the Meteorological Society of Japan.* Ser. II, 97(5), 931–965. https://doi.org/10.2151/jmsj.2019-051
- Yukimoto, S., Yoshimura, H., Hosaka, M., Sakami, T., Tsujino, H., Hirabara, M., et al. (2011). Meteorological research institute-earth system model version 1 (MRI-ESM1)—Model description (Vol. 64, pp. 1–96). Technical Reports of the Meteorological Research Institute.
- Zhang, R., Sutton, R., Danabasoglu, G., Kwon, Y.-O., Marsh, R., Yeager, S. G., et al. (2019). A review of the role of the Atlantic meridional overturning circulation in Atlantic multidecadal variability and associated climate impacts. *Reviews of Geophysics*, 57(2), 316–375. https:// doi.org/10.1029/2019RG000644
- Ziehn, T., Chamberlain, M. A., Law, R. M., Lenton, A., Bodman, R. W., Dix, M., et al. (2020). The Australian Earth system model: ACCESS-ESM1.5. *Journal of Southern Hemisphere Earth Systems Science*, 70(1), 193–214. https://doi.org/10.1071/ES19035